Author's response to reviews

Title: A new strategy and its effect on adherence to intermittent preventive treatment of malaria in pregnancy in Uganda.

Authors:

Anthony K Mbonye Prof (vpadmin@infocom.co.ug)
Josephine Birungi Dr (jbirungi@yahoo.com)
Stephanie Yanow Dr (staphanie@yahoo.com)
Pascal Magnussen Prof (pmla@life.ku.dk)

Version: 2 Date: 11 January 2013

Author's response to reviews:

Dear Editor

Thanks for sending back the manuscript for editing, I have revised it as follows:
1. The references in the text have been put in the right format
2. The references have been edited in the reference section
3. The ethics section has been put at the end of the methods section

Anthony